Author’s response to reviews

Title: Triglycerides and total Cholesterol concentrations in Association with IFG/IGT in Chinese Adults in Qingdao, China

Authors:
zulqarnain baloch (znbalooch@yahoo.com)
jing cui (Cuijing_0623@163.com)
Jianping sun (qdcdcsjp@126.com)
wang wei (657536523@qq.com)
Nafeesa Yasmeen (zeshanfeed@gmail.com)
Ma Ke (871667400@qq.com)
Hualei Xin (Xinhualei1987@126.com)
Qing Qiao (116661036@qq.com)
Aiguo Ma (864553361@qq.com)

Version: 1 Date: 25 Oct 2017

Author’s response to reviews:

Dear Sir,

We greatly appreciate your efforts and consideration of our manuscript for possible publication. I have revised this manuscript carefully according to his suggestions. Detailed responses to the comments and questions are provided below.

comment 1:

1. We note that the email address of the author Qing Qiao mentioned in the manuscript is different from that mentioned in the editorial submission system. Please ensure that the email address is identical in the manuscript and in the editorial submission form.

Response: i edit the correct email ID of Prof. Qing Qiao
comment 2:

We note that the affiliation of the corresponding author Zulqarnain Baloch mentioned in the manuscript is different from that mentioned in the editorial submission system. Please ensure that the affiliation is identical in the manuscript and in the editorial submission form.

Response: sorry for this mistake. I edit my affiliation name in editorial submission system.

comment 3:

Please amend your sub-heading ‘Ethical statement’ to ‘Ethics and consent to participate’ within your manuscript.

Response: I edit the subheading as per your recommendation.

comment 4: Thank you for stating "ethical approval was obtained from Qingdao Municipal Center for Disease Control and Prevention". Please amend your current ethics statement to confirm that the Qingdao Municipal Center for Disease Control and Prevention has an ethics committee that approved your study.

Response: I revise the detail about ethical approval as per your recommendation.

comment 5: Please include a 'Consent for Publication' section in the Declarations as detailed here (https://bmcpublichealth.biomedcentral.com/submission-guidelines/preparing-your-manuscript/research-article). If this Declarations sub-heading is not relevant to your study design or article type, please write "Not applicable" in this section.

Response: I add the Consent for Publication heading in the revisions submission.

comment 6:

We note that authors Nafeesa Yasmeen and Ma Ke listed in the manuscript do not appear in the 'Author Contributions' section of the manuscript. Please amend your Author Contributions to
clearly state the involvement of all authors. Please see our editorial guidelines for Authorship at https://www.biomedcentral.com/getpublished/editorial-policies#authorship

Please also add a statement to the “Authors’ contributions” confirming that all authors read and approved the final manuscript.

Response: I add the contribution of Ma Ke and Nafeesa Yasmen in the author contribution section. Further, I also add that "all authors read and approved the final manuscript."

Comment 7: Please include a statement regarding Ethics and Consent in the methods section as well.

Response: Iby following your advice I add the a statement regarding Ethics and Consent in the methods section as well.

I revised my manuscript according to your recommendation. However, if still there is any mistake please inform me I will make it correct. Thank you again for your attention on these revisions. I look forward your decision.

Sincerely Yours,

Zulqarnain Baloch

College of veterinary medicine,

South China Agricultural University, China

Email: znbalooch@yahoo.com

Tel: 86-18344564625